% !TeX root = ../easyfloats.tex

\addtocontents{toc}{\protect\setcounter{tocdepth}{1}}

\section{Change log}
The Gitlab symbols are links to the corresponding commit/tag on Gitlab.

\begin{changelog}
\changesinversion{v1.1.0}

New features:
- `placement = I`
  \commit 05d13be6b7a6c79eb6fc2c7625372a8068d6dab9
- `\splitobject`
  \commit 19df2f6ba1565b7b8d1685410937c6d3cd7d22a8
- `subpage outer pos = auto-inverted|Auto-inverted`
  \commit 79ccf8d6b9e66f4629059fab4b93ff32f49ddfba
- `auto label prefix`
  \commit 0e59759ba22e26a081f5c943efdab4325593896e
- `\AtEndObject`
  \commit 2a38c33717406d620eb79b51f35873e50acd220c

Bugfixes:
- close all groups which `\includegraphicobject` opens
  \commit 52783c7ca6a7ed89f4c2610e18996c5737a81274
- don't indent subobjects on 2nd/3rd/xth line
  \commit 1793809c2430b1c35c5b51b5bdd5ecdc23202075

Documentation additions:
- new example: custom options, \cref{custom-options}
  \commit d6a2818d0405a5dd8e5fd581a2ae9f7588289064
- new example: \pkg{varioref}, \cref{vref}
  \commit f6754fe70e90aa81c201ec58fe31bf416a47efa9
- new example: subfigures, \cref{exp:subfigures}
  \commit ce6e49151630f74815da161e1df718092dfce9b9
- new example: how to create new float type, \cref{exp:new-object-type}
  \commit 73942fff3a4c3cf385c300e25efc0275083b68f5
- new example: how to circumvent subcaptionbox limitations, \cref{exp:subcaptionbox-limitations}
  \commit 96ae47cbab2d1d3185f2acb8ae9ca91f012d8e5e
- added more background information to the first example
  \commit 573f5116795fa9e2211da54ed909a04055b8ccc2
- how to achieve a different graphics width for subfigures
  \commit b2d1d65696675056fa35aaf8a83827b99fa2d7f3
- revised subobjects example
  \commit b5d368f11141e7c35cded3c10be8836d58073012
- added note which LaTeX version has been used for testing
  \commit d1d412bf29f3f30229c4c3b6bab38865bf39421f
- added more info to examples
  \commit b3b2de787075ef87da431f91c5a6f0dfb7eee8f1
- mention TikZ library perspective
  \commit 7b7dcda1656af9d5467e5d179ba367066890d51c
- `\crefname`
  \commit 9b6c609349235bf5e0779beb1eceea5f9067a7cb
- `\Crefname`
  \commit e0f7c551313e606e94b8b969386fb197f90a5683
- show texdoc command to open the corresponding document in \cref{references}
  \commit 193e5bf28b9d5fb747a74f9c049698fdabfadb27

Documentation updates:
- updated section regarding the installation
  \commit 463f9c01ce17d45e64fe0f028216b1cace7a7ab5
- `\floatplacement{figure}{H}` works nowadays
  \commit 5a76cc9cccec893ccab651bcc3f263906d237120

Documentation fixes:
- fixed wrong example, \cref{new-object-style-tikzobject}
  \commit bd9c60e89b8d2bf4b4c80e3d855261e1a53e9624
- added initial value of `table break text` to documentation
  \commit 4e9f3eb8e843e9826111882d1c30096652f36c40
- table break text is an executed key, not a storing key
  \commit c0eb27b32d1bae315a1d8c921fba1a0233411f8e
- fixed typos
  \commit 63d9bdcf40487a61e666045f449ff5cd8daeedd6
  \commit 1b22b422b8a9b816f8ce67050b28e3ccd9121260
- cleveref noabbrev: reworded to make it a neutral statement
  \commit 556fb4888a52b5b53885a8e353d29fc5d3e152ab
- show correct tilde symbols
  \commit b4ba76f1d5144c6d9c4002abcd2640a1077259d7
- updated comment on \pkg{lmodern}
  \commit 6d62a4b0768bc16df1deeccb4c2423fff7528459
- set label in first two examples instead of suppressing the warning
  \commit 5b84ceed5afba69081569a460b726091ebd28fae
- added prefixes to labels in examples
  \commit 54a1459d8c73acaf24464411341197a914198a5f

\changesinversion{v1.0.0}
The first release.
\end{changelog}
